

Reply to reviewers' comments

Sir

Thank you very much for your valuable comments. We modified manuscript as per your suggestion.

Thank you very much

Regards

Raveendran

## REPLY TO SPECIFIC COMMENTS TO AUTHORS

1. Reference numbers in the main text. The reference does not require superscripts, please modify it. The correct format is "the spleen hilum[3]".

SIR

THANK YOU VERY MUCH SIR FOR YOUR VALUABLE SUGGESTIONS.  
REFERENCE MODIFIED ACCORDINGLY

2. I insist that is important o define your criteria for include or exclude any references. If you believes that your criteria is blood glucose values should be below the diagnostic cut off for pre diabetes- that means within normal limit and if it is below ADA cut off for diagnosis of diabetes but above pre-diabetes ranges, it is considered partial remission it is important this criteria is describe in your methodology

SIR

CRITERIA DESCRIBED IN METHODOLOGY AND ADA'S CRITERIA FOR THE  
DIAGNOSIS OF DIABETES CITED

---

Reviewer #1:

**Scientific Quality:** Grade C (Good)

**Language Quality:** Grade B (Minor language polishing)

**Conclusion:** Major revision

**Specific Comments to Authors:**

Dear authors Congratulations by your paper. We would like which criteria were used to define your references. Is important to describe which inclusion and exclusion criteria were adopted. What guidelines you used to define your references? To publish in World Journal of Methodology is expected that one guideline was used to based in submitted papers

SIR

THANK YOU FOR YOUR VALUABLE SUGGESTIONS  
IN THIS ARTICLE WE ARE TRYING TO PROPOSE CRITERIA FOR REMISSION OF DIABETES. TO DEFINE COMPLETE REMISSION BLOOD SUGAR/ HBA1C VALUE SHOULD BE BELOW THE DIAGNOSTIC CUT OFF FOR PRE- DIABETES- THAT MEANS WITHIN NORMAL LIMIT. IF IT IS BELOW ADA CUT OFF FOR DIAGNOSIS OF DIABETES BUT ABOVE PRE- DIABETES RANGES, IT IS CONSIDERED PARTIAL REMISSION. REQUIREMENTS FOR DIAGNOSING REMISSION IS NOT MENTIONED CLEARLY IN THE AVAILABLE LITERATURE. SO BASED ON AVAILABLE DATA AND PUBLISHED LITERATURE WE ARE PROPOSING CRITERIA, WHICH WILL BE USEFUL FOR HEALTHCARE PROFESSIONALS WORLDWIDE

**Editorial Office's comments.** Authors must revise the manuscript according to the Editorial Office's comments and suggestions, which are provided below:

(1) *Science Editor:*

**1 Scientific quality:** The authors submitted a review of remission of type 2 diabetes mellitus.

(1) **Classification:** Grade C;

(2) **Summary of the Peer-Review Report:** It is important to describe the inclusion and exclusion criteria used. What criteria does the author use to define references? To be published in the *World Journal of Methodology*, a guideline is expected to be used in the submitted paper;

(3) **References recommendations:** The reviewer didn't request the authors to cite improper references published by him/herself.

(4) **Manuscript Type:** After verification, the manuscript type is "Minireviews".

## 2 Specific comments

(1) **Country/Territory of origin:** Please verify if the "Country/Territory of origin: India" submitted by the system is correct?

(2) The language classification is Grade B. Please visit the following website for the professional English language editing companies that we recommend:

<https://www.wjgnet.com/bpg/gerinfo/240>.

(3) **Please add the Running Title:** A short running title of no more than 6 words should be provided. Abbreviations are permitted. **For example,** Losurdo G *et al.* Two-year follow-up of HCC.

(4) **Please add the Author contributions:** The 'Author contributions' passage describes the specific contribution(s) made by each author. The author's names will be listed in the following format: full family (sur)name, followed by abbreviated first and middle names. For example, Bryan L Copple should be revised as Copple BL. A full multi-author example is: Wang CL, Liang L, Fu JF, Zou CC, Hong F and Wu XM designed the research study; Wang CL, Zou CC, Hong F and Wu XM performed the research; Xue JZ and Lu JR contributed new reagents and analytic tools; Wang CL, Liang L and Fu JF analyzed the data and wrote the manuscript; All authors have read and approved the final manuscript.

5) **The "Key Words" does not meet the requirements:** The 'Key words' list will provide 5-10 keywords that reflect the main content of the study. Please do not use abbreviations for the keywords (*e.g.*, Ulcerative colitis, not UC). The first letter of each keyword will be capitalized, and each keyword will be separated by a semicolon, with no terminal period. **An example of correct formatting is:** Non-alcoholic fatty liver disease; Alcoholic liver disease; Non-alcoholic steatohepatitis; Insulin resistance; Oxidative stress.

(6) **Reference numbers in the main text.** The reference does not require superscripts, please modify it. The correct format is "the spleen hilum[3]".

**If there are more than 3 consecutive references cited at the same time, use a "-" connection. For example,** This condition is defined in the absence of excessive alcohol intake, other liver diseases, or the consumption of steatogenic drugs[2-4]. **If there are only 2, use ",". For example,** mitigating the risk of disease progression[10,11].

The author should number the references in Arabic numerals according to the citation order in the text.

The format of in-text citation of references should be [References Number], which should be with no space between "[ ]" and the preceding word. **Example:** The pathophysiology is thought to be due to an increased arterial flow that leads to secondary hepatocellular hyperplasia[1,2].

If the name of the author(s) of a reference is listed in the sentence, the reference number should be placed immediately after the author(s) of the reference. **Example:** Mandal *et al*[8] proposed that retractor aponeurosis disinsertion is the most likely cause of congenital low lid entropion.

In addition, please verify the order and total number of references cited to ensure that all references in the list are cited and in a correct numeric order.

(7) **References.** To ensure the accuracy of the references, please use "Edit References by Auto-Analyser" (<https://www.f6publishing.com/Forms/main/ArticleReferenceTool.aspx>) to edit the references of the manuscript.

(8) **Figures.** *Authors should place figure legends, figures, in separate pages at the end of the manuscript.*

Figures must be presented in the order that they appear in the main text of the manuscript (numbered as 1, 2, 3, etc.). All figures must have a detailed figure legend that provides a clear and comprehensive description of the information presented in the figure, so that the reader can understand without having to refer back to any other portion of the manuscript. Uniform presentation should be used for figures showing the same or similar contents; for example, "**Figure 1 Pathological changes of atrophic gastritis after treatment.** A: ...; B: ...; C: ...; D: ...; E: ...; F: ...; G: ...".

Abbreviations must be defined upon first appearance in the **Figure Legends**. Do not use non-standard abbreviations, unless they appear at least two times in the text preceding the first usage/definition.

**Original figure documents.** *In the meantime, authors should provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor, and upload it to the file destination of "Image File" in the F6Publishing system.*

(9) **Tables.** *Authors should place Table Titles, Tables, and Table Notes in separate pages at the end of the manuscript.*

Tables must be presented in the order that they appear in the main text of the manuscript (numbered as 1, 2, 3, etc.). A brief, one-line title must be provided for each table.

Authors are required to provide standard three-line tables, that is, only the top line, bottom line, and column line are displayed, while other table lines are hidden. The contents of each cell in the table should conform to the editing specifications, and the lines of each row or column of the table should be aligned. Do not use carriage returns or spaces to replace lines or vertical lines and do not segment cell content.

The title "Table 1" or "Table 2" does not require a period to be added, and the title of the table does not have a period. For example, "**Table 2 Characteristics of the studies included in the meta-analysis**".

(10) Please obtain permission for the use of picture(s). If an author of a submission is re-using a figure or figures published elsewhere, or that is copyrighted, the author must provide documentation that the previous publisher or copyright holder has given permission for the figure to be re-published, and correctly indicate the reference source and copyrights. For example, "**Figure 1 Histopathologi**

**nation by hematoxylin-eosin staining (200 ×).** A: Control group; B: Model group; C: Pioglitazone hydrochloride group; D: Chinese herbal medicine group. **Citation:** Yang JM, Sun Y, Wang M, Zhang XL, Zhang SJ, Gao YS, Chen L, Wu MY, Zhou L, Zhou YM, Wang Y, Zheng FJ, Li YH. Regulatory effect of a Chinese herbal medicine formula on non-alcoholic fatty liver disease. *World J Gastroenterol* 2019; 25(34): 5105-5119. Copyright ©The Author(s) 2019. Published by Baishideng Publishing Group Inc<sup>[6]</sup>. And please cite the reference source in the references list. If the author fails to properly cite the published or copyrighted picture(s) or table(s) as described above, he/she will be subject to withdrawal of the article from BPG publications and may even be held liable.

**3 Recommendation:** Conditional acceptance.

Language Quality: Grade B (Minor language polishing)

Scientific Quality: Grade C (Good)

SIR

THANK YOU VERY MUCH SIR FOR YOUR VALUABLE SUGGESTIONS. REFERENCE, KEY WORD, TABLES AND OTHER RELEVANT AREA MODIFIED ACCORDING TO THE SUGGESTION. PPT FORMAT OF ALL THE FIGURES PROVIDED. GRAMMATIC AND TYPING ERRORS CORRECTED

*(2) Company Editor-in-Chief:*

I have reviewed the Peer-Review Report, full text of the manuscript, all of which have met the basic publishing requirements of the *World Journal of Methodology* and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office's comments and the Criteria for Manuscript Revision by Authors. The author(s) must submit the revised manuscript online through the *Intelligent Manuscript*

SIR

THANK YOU VERY MUCH SIR FOR YOUR VALUABLE SUGGESTIONS.

### **Special comments from the Editor:**

The revised manuscript does not conform to the journal's formatting and technical requirements.

Please re-revise your manuscript accordingly.

Special Comments:

- (1) Table 2 needs to be modified according to the specification: Authors are required to provide standard three-line tables, that is, only the top line, bottom line, and column line are displayed. while other table lines are hidden. The contents of each cell in the table should conform to the editing specifications, and the lines of each row or column of the table should be aligned. Do not use carriage returns or spaces to replace lines or vertical lines and do not segment cell content.

SIR

THANK YOU VERY MUCH SIR FOR YOUR VALUABLE SUGGESTIONS. TABLE 2  
MODIFIED ACCORDINGLY

---